Author’s response to reviews

Title: A de novo ANK1 mutation associated to hereditary spherocytosis: a case report

Authors:

tilong huang (tilonghuang@etyy.cn)
baohua sang (baohuasang@etyy.cn)
Qing-Ling Lei (qinglinglei@etyy.cn)
chunyan song (chunyansong@etyy.cn)
yunbi lin (yunbilin@etyy.cn)
yu lv (yulv@etyy.cn)
chunhui yang (chunhuiyang@etyy.cn)
nal i (nali@etyy.cn)
yuehuang yang (yuehuangyang@etyy.cn)
xianwen zhang (xianwen-zhang0717@163.com)
xin tian (xintian@etyy.con)

Version: 1 Date: 03 Sep 2018

Author’s response to reviews:

1. Please include 'Declarations' heading in your Declarations section.
   Answer: Yes

2. CARE guidelines
   Answer: Yes

Editor's comments:

1. While assessing your manuscript we found several instances where the text displayed overlap with other previously published works, in particular:
   Answer: we have referenced
2. In accordance with BioMed Central editorial policies, please ensure your manuscript adheres to CARE guidelines for reporting case reports.

Answer: Yes

3. In your Ethics approval and consent to participate subsection of the Declarations, please clarify whether the family of the proband gave their consent to participate in the genetic testing conducted upon them.

Answer: Yes